Author's response to reviews

Title: Ventricular-arterial uncoupling in heart failure with preserved ejection fraction after myocardial infarction in dogs - invasive versus echocardiographic evaluation.

Authors:

Myrielle Mathieu (myrielle.mathieu@ulb.ac.be)
Bachar El Oumeiri (beloumei@ulb.ac.be)
Karim Touihri (ktouihri@ulb.ac.be)
Ielham Hadad (ielham.hadad@ulb.ac.be)
Maryam Mahmoudabady (maryam.mamoudabadi@ulb.ac.be)
Philippe Thoma (phthoma@ulb.ac.be)
Thierry Metens (tmetens@ulb.ac.be)
Jozef Bartunek (jozef.bartunek@olvz-aalst.be)
Guy Heyndrickx (guy.heynrickx@olvz-aalst.be)
Serge Brimioulle (serge.brimioulle@ulb.ac.be)
Robert Naeije (rnaieje@ulb.ac.be)
Kathleen Mc Entee (kmcentee@ulb.ac.be)

Version: 4 Date: 28 June 2010

Author's response to reviews:

Dear Editor,

It is my pleasure to resubmit the manuscript as being ready for publication. We have made required changes to the tables and figure 1 as well as checked the text for typographical errors.

Sincerely yours
Kathleen Mc Entee